Author's response to reviews

Title: A study protocol to evaluate the relationship between outdoor air pollution and pregnancy outcomes

Authors:

Manuel C Ribeiro (manuel.ribeiro@ist.utl.pt)
Maria J Pereira (maria.pereira@ist.utl.pt)
Amílcar Soares (asoares@ist.utl.pt)
Cristina Branquinho (cmbranquinho@fc.ul.pt)
Sofia Augusto (s.augusto@fc.ul.pt)
Esteve Llop (ellv66@gmail.com)
Susana Fonseca (susanafonseca@yahoo.com)
Joaquim G Nave (joaquim.g.nave@gmail.com)
António B Tavares (antonio.tavares@insa.min-saude.pt)
Carlos M Dias (carlos.das@insa.min-saude.pt)
Ana Silva (ana.silva@alentejolitoral.min-saude.pt)
Ismael Selemane (ismael@csgrandola.min-saude.pt)
Joaquin de Toro (jdetoro@cssantiagoc.min-saude.pt)
Mário J Santos (lince@cssantiagoc.min-saude.pt)
Fernanda Santos (fernanda.santos@alentejolitoral.min-saude.pt)

Version: 5 Date: 12 October 2010

Author's response to reviews: see over
“A study protocol to evaluate the relationship between outdoor air pollution and pregnancy outcomes”, Manuel C Ribeiro et al.

Manuscript ID: 4545194733978541

To BMC Series Editorial Production Team

Please find enclosed the final version of manuscript.
We believe we have addressed all formatting changes, as discussed below on a point-by-point basis.

Major revisions

1. Authors' contributions: Please indicate that all authors read and approved the final manuscript. Done.

2. Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. We included the tables in a tables section following the references section in order to be included in the manuscript.

3. References: References should be cited in the text using consecutive numbers in square brackets. Done.

Minor revisions

4. Affiliations: Please remember to include the city and country in the affiliation details. Done.

5. Abstract: Please rename the section titled 'Methods' in the abstract 'Methods/Design'. Done.

6. Methods: Please rename this section 'Methods/Design'. Done.

7. References: the reference list should contain up to 30 authors' names for each citation. Done.

8. References: please provide the full citation details for reference 35. Done.

9. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. Done.

Regards,
On behalf of all co-authors
M C Ribeiro